Author's response to reviews

Title: Networking between community health programs: a team-work approach to improving health service provision

Authors:

Nathan J Grills (nathangrills@gmail.com)
Robert Kumar (robert@eha-health.org)
Maneesh Phillip (maneesh.phillip@gmail.com)
Gillian P Porter (gillianpporter@gmail.com)

Version: 2 Date: 12 August 2013

Author's response to reviews:

The Editor
BMC Health Services Research
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom
12 August 2013

RE: Revised article submission – Networking between community health programs: a team-work approach to improving health service provision

Dear Sir/Madam,

Thank you for considering our research paper ‘Networking between community health programs: a team-work approach to improving health service provision.’

In accordance with your requests we have revised the submitted article in the following ways:

1. Documented the consent process which was undertaken in the research process
2. Revised the conclusions, in particular we have stated the main conclusion of our research and discussed their implications
3. Reviewed the RATs guidelines – we have reread the RATs guidelines and believe our research article adheres to these.

Yours sincerely,

Nathan Grills